

# GOPEN ACCESS

**Citation:** Loeffel M, Ross A (2022) The relative impact of interventions on sympatric *Plasmodium vivax* and *Plasmodium falciparum* malaria: A systematic review. PLoS Negl Trop Dis 16(6): e0010541. https://doi.org/10.1371/journal. pntd.0010541

**Editor:** Rhoel Ramos Dinglasan, University of Florida, UNITED STATES

Received: February 13, 2022

Accepted: May 27, 2022

Published: June 29, 2022

**Peer Review History:** PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0010541

**Copyright:** © 2022 Loeffel, Ross. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** The data are in the Supporting Information files and at <a href="https://github.com/rossaman4/vivax-falciparum-relative-impact">https://github.com/rossaman4/vivax-falciparum-relative-impact</a>.

RESEARCH ARTICLE

# The relative impact of interventions on sympatric *Plasmodium vivax* and *Plasmodium falciparum* malaria: A systematic review

#### Melanie Loeffel<sup>1,2</sup>, Amanda Ross<sup>1,2</sup>\*

1 Swiss Tropical and Public Health Institute, Allschwil, Switzerland, 2 University of Basel, Basel, Switzerland

\* amanda.ross@unibas.ch

# Abstract

## Background

In areas with both *Plasmodium vivax* and *Plasmodium falciparum* malaria, interventions can reduce the burden of both species but the impact may vary due to their different biology. Knowing the expected relative impact on the two species over time for vector- and drugbased interventions, and the factors affecting this, could help plan and evaluate intervention strategies.

## Methods

For three interventions (treated bed nets (ITN), mass drug administration (MDA) and indoor residual spraying (IRS)), we identified studies providing information on the proportion of clinical illness and patent infections attributed to *P. vivax* over time using a literature search. The change in the proportion of malaria attributed to *P. vivax* up to two years since implementation was estimated using logistic regression accounting for clustering with random effects. Potential factors (intervention type, coverage, relapse pattern, transmission intensity, seasonality, initial proportion of *P. vivax* and round of intervention) were assessed.

#### Results

In total there were 55 studies found that led to 72 series of time-points for clinical case data and 69 series for patent infection data. The main reason of study exclusion was insufficient information on interventions. There was considerable variation in the proportion of malaria attributed to *P. vivax* over time by study and location for all of the interventions. Overall, there was an increase apart from MDA in the short-term. The potential factors could not be ruled in or out. Although not consistently significant, coverage, transmission intensity and relapse pattern are possible factors that explain some of the variation found.

## Conclusion

While there are reports of an increase in the proportion of malaria due to *P. vivax* following interventions in the long-term, there was substantial variation for the shorter time-scales considered in this study (up to 24 months for IRS and ITN, and up to six months for MDA).

**Funding:** The authors received no specific funding for this work.

**Competing interests:** The authors have declared that no competing interests exist.

The large variability points to the need for the monitoring of both species after an intervention. Studies should report intervention timing and characteristics to allow inclusion in systematic reviews.

#### Author summary

In areas with both *P. vivax* and *P. falciparum* malaria, an intervention may impact the species differently due to their different biology. In particular, an inoculation with P. vivax can lead to liver-stage parasites which relapse weeks or months later. The intervention may affect the species equally or one more than the other and this may change over time. Knowledge of the expected relative effect of interventions would be useful for tailoring intervention strategies to address both species, and for interpreting monitoring data. This study aims to assess patterns of the proportion of clinical cases or patent infections that were P. vivax or P. falciparum on a short time scale of up to two years since implementation and to identify potential factors that could explain variation in these observed patterns for two vector-based interventions (insecticide treated nets and indoor residual spraying) and a drug-based intervention (mass drug administration). In all, there were 72 series of time-points for clinical data and 69 for patent infections. Overall there was an increasing proportion of P. vivax over time, but there was substantial variation between studies for all interventions. Coverage, transmission and relapse pattern were identified as possible explanations for some of the variation, however, this remains uncertain due to the high variation and the data available.

#### Introduction

Of the species of malaria infecting humans, *Plasmodium falciparum* and *Plasmodium vivax* occur the most frequently and together accounted for 241 million cases in 2020 [1]. Separately they have different geographical ranges, but they co-exist particularly in the horn of Africa and Madagascar, South and Central America, the South-East Asian region and the Western Pacific region [2,3].

Both vector- and drug-based interventions can affect both species. In areas where both species are prevalent, the impact of the interventions may differ. Different patterns of the ratio of *P. vivax* to *P. falciparum* cases over time have been reported. One study in Thailand [4] with treatment of cases reported equal reductions in both species. Another in Papua New Guinea [5], where the usage of nets was increased found that *P. vivax* became dominant. A recent review by Price [6] found that control often leads to an increase in the ratio of malaria attributed to *P. vivax* over longer periods and describes studies where initial effects are a decrease in *P. falciparum* malaria followed by a decrease in *P. vivax* later [6–8] suggesting different dynamics over time.

Possible explanations for different relative effects can lie in biological differences, particularly the ability of *P. vivax* to form dormant hypnozoites in the liver which can relapse weeks or months later [9]. The relapses are likely to decrease the time to patent re-infection following clearance of blood-stage parasites by drugs, and to lower the impact of vector control on reducing blood-stage infections until the relapses have been depleted. Other possible factors are characteristics of the setting such as transmission intensity or seasonality, and characteristics of the intervention, such as the mode of action, coverage, and whether it is the first implementation or a continuation of previous implementations since the dynamics for these can differ [10].

Program managers and decision-makers need to know the likely impact of interventions in their settings. Knowledge of the expected relative impact over time and the drivers in different settings may also allow easier interpretation of data for evaluating the impact of an intervention strategy, allow the tailoring of combinations of interventions for the optimal impact on both species and inform epidemiology.

We investigate the relative impact of vector-control and drug-based interventions on *P*. *vivax* and *P*. *falciparum* on a short-term time scale up to two years after implementation. We carry out a literature search to identify studies in areas with both *P*. *vivax* and *P*. *falciparum* which have documented distributions of treated bed nets (ITN), indoor residual spraying (IRS) or mass drug administration (MDA). We then estimate the variation in the observed trends of the proportions of clinical cases and patent infections that are attributed to *P*. *vivax* after the implementation of the interventions. We also aim to identify factors that affect the change in the proportion of clinical cases or patent infections.

#### Methods

#### Literature search

Identifying potential studies. The search terms for the PubMed search were: (("falciparum"[tiab] OR "Malaria, Falciparum"[Mesh] OR "Plasmodium, Falciparum"[Mesh]) AND ("vivax"[tiab] OR "Malaria, Vivax"[Mesh] OR "Plasmodium, Vivax"[Mesh])) AND ("Insecticides/administration and dosage"[Mesh] OR "Mass Drug Administration"[Mesh] OR "Insecticide-Treated Bednets"[Mesh] OR "IRS"[tiab] OR "indoor residual spraying"[tiab] OR "Prevalence"[Mesh] OR "Incidence"[Mesh]). This search in PubMed generated 973 hits (03.08.2020). Published reviews were also screened. Cochrane reviews on mass drug administration (MDA) [11], indoor residual spraying (IRS) [12], insecticide treated nets (ITNs) [13,14] and IRS and ITN [15] were screened for studies that took place in areas with both malaria species. Furthermore, if papers that were found with the other two approaches cited papers that could potentially be useful, then these were included for screening as well.

Papers identified as potentially relevant were imported into Endnote for abstract screening. Studies that were not in English or German were excluded. Studies that based on the abstract did not fulfill the inclusion criteria (Box 1) were not considered potentially relevant. If relevance was unclear, they would be included in the full-text screening. In particular, the interventions implemented were frequently unclear based on the abstract alone.

#### Box 1

#### Inclusion criteria

- 1. Report both P. vivax and P. falciparum in the same area
- 2. Report incidence and/or prevalence on at least two time-points or have a concurrent control group
- 3. Intervention in between two surveys or control groups that did not receive the intervention

#### Full-text screening of papers identified as potentially relevant

The papers considered as potentially relevant based on the abstract underwent full-text screening to see if they fulfilled the inclusion criteria (Box 1). This was carried out by one reviewer (ML) and discussed in detail with a second reviewer (AR).

Some studies would fulfill these criteria, however, needed to be excluded due to various reasons (Box 2).

Box 2

#### Exclusion criteria

- 1. Insufficient information about the intervention
- 2. Only one time-point measured
- 3. Epidemic malaria
- 4. Cases of one species being mainly due to imported cases
- 5. Population consisting solely of pregnant women

The information on the timing and coverage of interventions needed to be sufficiently detailed. For all three interventions, in most cases the starting point could be pinpointed to the month. For IRS and MDA the starting point needed to be within a window of not more than two years but the interventions could happen over longer time periods. In case of treated bed nets there needed to be a distribution (not just the availability of treated bed nets) with the time-point of the distribution within a window of two years.

Some studies would report having malaria control programs but did not indicate information on timing or control interventions and therefore were excluded. In other cases, there were different interventions implemented at the same time which would not allow disentangling the impacts and therefore were excluded. Interventions were included no matter if it was the first time they happened or a repeated time, but this was noted in order to be able to distinguish between first or repeated rounds of intervention.

Further interventions such as repellents, prophylaxis and larviciding were beyond the scope of this review and were excluded. However, studies were included if they did such interventions as well as those of interest and a note was taken of the additional intervention. Randomized control studies could be included if either they reported measurements before the intervention was implemented as well or if they provided a control group.

# Data extracted from the papers and additional variables from other sources

Extracted information was transferred into a Microsoft Access database. Several manuscripts could be combined if they reported on the same area to give one series of time-points. Similarly, if one study reported different areas or different age groups this would be added into the database separately leading to several series of time-points from one study.

#### **Outcome variables**

The number of clinical cases and prevalence of patent infections were extracted from the studies. If the studies did not include incidence or prevalence as a table or in the text but had a figure the web plot digitizer [16] was used to extract the values of the data points.

#### Rationale for selection of variables

**Intervention and round of intervention.** Drug- and vector-based interventions have different effects over time. For MDA, the initial impact occurs quickly by clearing blood-stage infections. However, MDA does not protect against new blood-stage infections beyond the prophylactic period of the drugs [17]. Vector-based interventions do not clear existing blood-stage infections but reduce the number of new infections from infected bites. For both vector-and drug-based interventions, new blood-stage infections can come from infected bites for both species and additionally, unless the hypnozoites are cleared or used up, *P. vivax* relapses. As a consequence, the proportion of cases attributed to *P. vivax* may increase in the short-term.

A modelling study on *P. falciparum* found that when nets are implemented for the first time, a drop in transmission is expected, whereas repeated rounds are expected to sustain the previous reduction or led to an increase in transmission via changes in the levels of acquired immunity in the community [10]. Therefore, the dynamics can differ according to round of intervention.

**Coverage.** If an intervention does affect one species more than the other, we would expect increased coverage to lead to a stronger change in the proportion.

**Transmission intensity.** Modelling studies have predicted stronger effects of ITN [10,18] and MDA [19] for *P. falciparum* in low transmission settings. The transmission intensities of *P. falciparum* and *P. vivax* are frequently different in the same setting.

Age. The age at the peak clinical incidence can differ between species in a setting, due both to their transmission intensities and the rate at which acquired immunity is gained for the species. As a consequence, interventions may lead to different impacts for the species.

**Relapse pattern.** The number and timing of the relapses per infected bite are likely to influence the timing of any changes in the proportion of cases that are *P. vivax*.

**Seasonality.** The effect of an intervention may vary depending on how seasonal the setting is, and the degree of seasonality can differ by species.

Season when the data was collected. The proportion of *P. vivax* to *P. falciparum* differs by the season that the measurement was taken in [20-22] due to the ability of *P. vivax* to relapse.

#### Definitions of variables

**Intervention and round of intervention.** The round of intervention was noted for MDA and ITN. The spraying rounds of IRS were not added individually but the starting point was noted. It was often not possible to distinguish between first and later rounds for IRS, since spraying in the previous year was not reported. Spraying was classed as repeated if spraying the year before was reported. In this case, the studies usually looked at a change in the insecticide used.

For ITN distribution, we assumed that if no previous distribution was mentioned then none had happened. An exception to this was a study by Deressa [23] where 65% of people already owned nets. Therefore, it was assumed there had been a previous distribution although not specifically mentioned. There was no distinction made between insecticide treated nets (ITNs) and long lasting insecticidal nets (LLINs). Studies were excluded when the previous intervention was likely to still have an effect: interventions were considered to not influence the effect of the new intervention substantially if the previous intervention was started more than six months before the intervention of interest, if the concurrent intervention was an increase in case management, or if the study gave reasons to believe that the previous intervention was not very impactful.

**Coverage.** Studies were categorized into higher or lower coverage. An arbitrary cut-off of 70% was used for ITNs based on an equal division of the studies.

The studies had various ways of quantifying ITN coverage. The preferred metric of coverage for bed nets was usage followed by the metrics listed in Box 3. In many cases only one of these measures was available.

Box 3

#### Quantifying the Coverage of ITN

- 1. Usage of bed nets
- 2. Coverage of bed nets
- 3. Calculation using number of nets and population assuming 1 net covers 2 people
- 4. If usage in different seasons is given, it was assumed to be high if majority of seasons had usage above 70%
- 5. If self-reported usage was given for several years after distribution, first self-reported usage was considered
- 6. With several locations but coverage information for one only, it was assumed to be the same for other locations too.

For one study (Deressa [23]) the aim was to cover with one net per two people. It was not clear if this was reached but assumed it was and therefore the coverage would be high. In other cases, there were studies (Luxemburger [24], Rowland [25], Rowland [26]) where only a fraction of the community would receive nets and only those people would be tested. However, as only the people who received nets were tested the coverage was defined on what proportion of these people received a net. We recognize that community effects of the nets would be reduced in this situation.

As MDA and IRS generally had higher coverages, the cut off was set to 90% (of people receiving a dose or of households being sprayed). If there were several rounds of MDA coverage was considered high if 90% or more received at least one round.

The number of months since the intervention was implemented. The studies usually gave a time-span of the implementation of the intervention, which could be the year of implementation but was usually more exact dates. To define when the intervention was implemented (time zero), the mid-point of these time spans was taken. Similarly, the mid-point of time windows given for prevalence surveys and incidence data collection were used. In most series of MDA, there were several rounds over three months. The mid-point of the time span of the first round was used as the time of implementation.

The proportion of *P. vivax* cases just before the intervention was implemented is needed to allow the change in proportion over time to be estimated. To determine the proportion at this time-point, the closest time-point before the intervention was used unless there were several data points within six months before the intervention, in which case all of these were used in order to to provide a more stable estimate of the proportion.

However, some studies had measured malaria burdens more than six months before the intervention, these were included using the closest measurement before the intervention was used to avoid excluding many studies (for clinical case data: 15 series of time points (21%) (median time point of measurement of these series: 15 months before intervention), for patent infection data: 32 series of time points (46%) (median time point of measurement of these series: 30 months before intervention)). We assumed that if there had been no major interventions, then the proportion would be similar after accounting for season of measurement in the regression analyses. The median time of the pre-implementation points was 0.2 months for clinical cases and 9.7 months for patent infections. The data-points after the intervention was implemented were used up to the time of another implementation. This means that a datapoint could be used twice, once as a measurement after an intervention happened and once before a different intervention. In all the analyses, the season of the measurement was accounted for.

**Transmission.** The transmission intensities of *P. falciparum* and *P. vivax* frequently differ in a setting. *P. vivax* and *P. falciparum* were categorized separately into high and low transmission intensities which led to four transmission categories. The transmission intensity in the study settings was determined using malaria burden measures from the studies, however, the studies had non-identical measures of the incidence of clinical cases and prevalence. For example, incidence data could be reported in different age groups or collected actively or only passively. However as only a rough distinction between high and low transmission intensity was required, the information in the studies was used. If there were entomological inoculation rates given or prevalence these were used. An arbitrary cut off of 5% prevalence of *P. falciparum* and *P. vivax* was used to allow reasonable numbers of studies in the transmission categories.

When prevalence data was not available then incidence data was used. A transmission setting was considered high with more than 100 cases per 1000 people per year. In uncertain cases, other literature from the setting or maps from the Malaria Atlas Project would be consulted [2,3]. However, these maps are recent and only go back to 2000 while some of the studies were carried out in previous decades and therefore these maps could not be used for all studies.

**Relapse pattern.** To define the relapse pattern of *P. vivax* occurring in the study areas, the classification by White [27] was used. White distinguishes between long latency relapse patterns (relapses approximately after 8–10 months) and frequent relapse patterns with relapses shortly after the primary infection. The classification is based on the broad geographical regions in Fig 17 in White [27]: regions may have frequent relapse strains only, long latency strains only or both. Latency is also associated with the number of relapses with temperate strains displaying fewer relapses than tropical strains [27].

**Seasonality.** Seasonality of rainfall was used as a proxy for the seasonality of malaria transmission [28]. Although this is not the only factor that influences seasonality, it was readily available for all areas. The relative entropy quantifies how much the distribution of rainfall differs from a uniform distribution of the rain across all months of the year [29]. To assign seasonality, the model based on the Climate Prediction Center Merged Analysis of Precipitation dataset (CMAP model) of the relative entropy [29] was used. The threshold chosen to distinguish between low and high seasonality was an arbitrary value of 0.6.

**Seasonal patterns.** The season at the time-point of measurement was often reported in the studies. When it was missing other studies with information on the timing of seasons in the area were used. If the prevalence or incidence was an average over the year, the season of survey was assumed to be in both wet and dry seasons. If the season the data was collected in could not be identified, it was noted as missing.

**Sample size of studies.** The actual case numbers or number of people infected were extracted.

**Proportion of malaria attributed to P. vivax.** To calculate the proportion of malaria cases with *P. vivax* infection, the number of clinical cases (or patent infections) with *P. vivax* infection was divided by the number of clinical cases (or patent infections) with either *P. vivax* or *P. falciparum*. If the studies counted mixed cases separately, they were added to both *P. vivax* case numbers as well as *P. falciparum* case numbers, as this was mostly reported to have been done in the studies when the mixed case numbers were not given separately. There were studies where mixed cases counted towards one species only, this was noted in Table A-J in S1 Supporting Information. Only malaria cases with *P. vivax* and/or *P. falciparum* were used in the analysis.

#### **Risk of bias**

The risk of bias was assessed in general rather than for each individual study. None of the studies aimed to assess the relative impact of an intervention on *P. vivax* and *P. falciparum* directly, and so the details which may affect the risk of bias were generally not reported. We considered the seven domains of the ROBINS-I tool (confounding, selection of participants, bias in classification of an intervention, deviation from intended interventions, missing data, bias in measurement, selective reporting depending on findings) [30].

#### Statistical analysis

The analysis aimed to (i) describe changes in the proportion of cases attributed to *P. vivax* by time since the implementation of the intervention and (ii) to identify factors that influence the pattern. Separate analyses were carried out for clinical cases and patent infections. R (version 4.0.1) [31] was used.

To display the data visually, the proportion attributed to *P. vivax* was plotted over time for each series individually (number of *P. vivax* cases divided by number of *P. vivax* and *P. falcipa-rum* cases combined). The size of the point increased with higher number of cases found at that time-point. Spaghetti plots by factor gave a visual impression of the potential role of the factor.

Characteristics of the series of time-points, such as age-group, were summarized using proportions. R-packages used for data processing and the visualization of data were diagrammeR [32], maps [33], ggplot2 [34], rnaturalearth [35], plyr [36], reshape2 [37].

Logistic regression was used for the analysis with the proportion of either clinical cases or patent infections that were *P. vivax* as the outcome. The change in the proportion of *P. vivax* over time was described as the change in odds per month since implementation. The model includes random effects for both intercept and slope to account for clustering by series. Time was used as a continuous variable. Potential explanatory variables were included both as the variable alone and as an interaction with time, to assess their effect on the change in proportion over time.

Maximum likelihood was used to fit the models. The lme4 package [38] in R was used. In all models, the season of the survey was adjusted for since season is likely to affect the proportion of *P. vivax* to *P. falciparum*.



#### Fig 1. Flow diagram of studies in the review.

https://doi.org/10.1371/journal.pntd.0010541.g001

Randomized controlled studies with control groups were included. If no measurements before the intervention were available, it was assumed that before the intervention the proportion of *P. vivax* to *P. falciparum* was equal to the proportion in the control group that did not receive the intervention. This is equivalent to assuming the effect observed in the study would also have been observed longitudinally.

#### **Ethics statement**

This study did not require any special ethical clearance. No new data were collected: the data included in the review were extracted from published papers.

#### **Results**

#### Literature search

The literature search lead to 1157 hits of which 55 studies were included (Fig 1). The PubMed search yielded 973 hits, and 184 were obtained by screening citations in Cochrane Reviews and the included papers.

The full-text screening included 415 studies. The most common reason for exclusion was the lack of information on interventions (Table 1), either no interventions were reported, or it was not clear at what time-point interventions happened or what the intervention was. In 56 cases there were studies that either only did a survey at one time-point or if there were surveys at several time-points the results would be an average of these surveys rather than reporting

Table 1. Rea	sons for	exclusion.
--------------	----------	------------

Reason for exclusion	Number of studies		
Lacks information on intervention or no intervention	208		
No distinction of different time-points of the surveys	56		
No distinction of species across time	52		
One species not detected or not reported	13		
Imprecise because data is for the whole country	7		
Same data as another study that is already used	6		
Not outcome of interest	6		
Looks at epidemics	4		
Issues with disentangling intervention effect	5		
Very low prevalence of <i>P. falciparum</i>	1		
Excluded relapses of <i>vivax</i> from cases	1		
Imported cases	1		
Total	360		

https://doi.org/10.1371/journal.pntd.0010541.t001





them separately. In a further five studies it was not possible to disentangle the effect of the interventions because there were several interventions happened at the same time.

A broad description of the included studies can be found in <u>S1 Supporting Information</u>. The largest number of studies was in Papua New Guinea (nine studies, 20%) followed by India (eight studies, 17%). The remaining studies were located in Nicaragua, Venezuela, Brazil, Ethiopia, Palestine, Iran, Pakistan, Afghanistan, Cambodia, Vietnam, Myanmar, Thailand, Indonesia, Vanuatu and the Solomon Islands (Fig 2).

Some studies reported several interventions (not at the same time), or several locations, and could therefore provide multiple series of time-points. Others were combined since they reported on the same area. In total, there were 141 series of time-points, 69 of patent infections and 72 of clinical cases. For three ITN distributions [39–41], the round of intervention was unclear or another intervention was phased out as the distribution took place and therefore they were excluded from the analysis.

#### Summary of the data

Twenty-seven studies met the criteria to be included in the analyses of clinical cases, providing 72 series of time-points for different age-groups and locations and a total of 500 time-points. Overall, 16 (22%) of the series included IRS, 39 (54%) treated nets, and 17 (24%) drug treatment (Table 2). The study dates ranged from 1959 to 2019 with 21 series of time-points from studies before the year 2000 (range ITN: 1996–2019, range MDA: 1983–2019, range IRS: 1959–2019).

The majority of the series of time-points included all age-groups, with only two series focusing on under 5-year-olds and two series on under 10-year-olds (<u>Table 2</u> and <u>S1 Supporting</u> <u>Information</u>). Most of the series (92%) were observational and compared the clinical cases before and after the intervention was implemented. For six series (8%) there were control

	ITN (first distribution) (n = 26)	ITN (repeated distribution) (n = 13)	MDA (first round) (n = 13)	MDA (repeated round) (n = 4)	IRS (first spraying) (n = 4)	IRS (repeated spraying) (n = 3)	IRS (undetermined round) (n = 9)	Overall (n = 72)
Age group								
0-5 years	1 (4%)	1 (8%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	2 (3%)
0–10 years	0 (0%)	0 (0%)	0 (0%)	0 (0%)	1 (25%)	1 (33%)	0 (0%)	2 (3%)
10+ years	0 (0%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	1 (33%)	0 (0%)	1 (1%)
all ages	25 (96%)	12 (92%)	13 (100%)	4 (100%)	3 (75%)	1 (33%)	9 (100%)	67 (93%)
Coverage <sup>1</sup>			-			-		
low	9 (35%)	4 (31%)	10 (77%)	4 (100%)	0 (0%)	0 (0%)	0 (0%)	27 (38%)
high	15 (58%)	6 (46%)	2 (15%)	0 (0%)	1 (25%)	3 (100%)	7 (78%)	34 (47%)
missing	2 (8%)	3 (23%)	1 (8%)	0 (0%)	3 (75%)	0 (0%)	2 (22%)	11 (15%)
Seasonality <sup>2</sup>			-			-		
low	18 (69%)	0 (0%)	11 (85%)	4 (100%)	0 (0%)	3 (100%)	4 (44%)	40 (56%)
high	8 (31%)	13 (100%)	2 (15%)	0 (0%)	4 (100%)	0 (0%)	5 (56%)	32 (44%)
Relapse pattern			-			-		
frequent	13 (50%)	10 (77%)	3 (23%)	0 (0%)	0 (0%)	0 (0%)	1 (11%)	27 (38%)
long	4 (15%)	0 (0%)	9 (69%)	4 (100%)	2 (50%)	3 (100%)	3 (33%)	25 (35%)
both	9 (35%)	3 (23%)	1 (8%)	0 (0%)	2 (50%)	0 (0%)	5 (56%)	20 (28%)
Transmission <sup>3</sup>			-			-		
low Pf low Pv	15 (58%)	9 (69%)	9 (69%)	4 (100%)	3 (75%)	1 (33%)	5 (56%)	46 (64%)
low Pf high Pv	3 (12%)	0 (0%)	2 (15%)	0 (0%)	0 (0%)	0 (0%)	2 (22%)	7 (10%)
high Pf low Pv	6 (23%)	3 (23%)	0 (0%)	0 (0%)	1 (25%)	2 (67%)	0 (0%)	12 (17%)
high Pf high Pv	2 (8%)	1 (8%)	2 (15%)	0 (0%)	0 (0%)	0 (0%)	2 (22%)	7 (10%)
Diagnostic tool								
RDT	1 (4%)	0 (0%)	1 (8%)	0 (0%)	1 (25%)	0 (0%)	0 (0%)	3 (4%)
RDT or microscopy	10 (38%)	0 (0%)	2 (15%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	12 (17%)
RDT then microscopy	0 (0%)	7 (54%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	7 (10%)
microscopy	8 (31%)	4 (31%)	10 (77%)	4 (100%)	3 (75%)	3 (100%)	8 (89%)	40 (56%)
not reported	7 (27%)	2 (15%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	1 (11%)	10 (14%)
Type of study								
observational	23 (88%)	13 (100%)	13 (100%)	4 (100%)	3 (75%)	3 (100%)	7 (78%)	66 (92%)
trial	3 (12%)	0 (0%)	0 (0%)	0 (0%)	1 (25%)	0 (0%)	2 (22%)	6 (8%)
Survey taken in	different seasons <sup>4</sup>							
yes	6 (23%)	2 (15%)	12 (92%)	4 (100%)	1 (25%)	2 (67%)	7 (78%)	34 (47%)
no	20 (77%)	11 (85%)	1 (8%)	0 (0%)	3 (75%)	1 (33%)	2 (22%)	38 (53%)
Time points per	time series							
mean	4.3	5.8	14	6.3	6.8	4.7	7.2	6.9
median (min, max)	5 (2,9)	3 (2,24)	12 (5,34)	6 (6,8)	3.5 (2,18)	5 (4,5)	4 (2,15)	5 (2,34)

#### Table 2. Characteristics of the 72 series of time-points of clinical cases by intervention.

<sup>1</sup>High coverage is defined as more than 70% of people with access to intervention for ITN and 90% of people for IRS. For MDA high coverage is defined as more than 90% receiving at least one treatment

<sup>2</sup>Seasonality is defined using the measure of relative entropy described by Pascale (2015) [29] which quantifies the distribution of rainfall across the year

<sup>3</sup>Transmission intensity cut off was at 5% prevalence for the respective species or less than 100 cases per 1000 people per year

<sup>4</sup>The data collection may spread across different seasons or all the data points described may be collected throughout the same season (e.g. always during the wet season)

https://doi.org/10.1371/journal.pntd.0010541.t002

groups as comparison instead of clinical cases before the intervention was implemented. In more than half of the series of time-points microscopy was used as the diagnostic tool. Most series were set in areas with low transmission for both *P. vivax* and *P. falciparum*. The total number of clinical cases found per time-point ranged from 1 to 19020 cases.

There were 24 studies included with information on the prevalence of patent infections (Table 3), leading to a total of 69 series of time-points and 198 individual time-points. There were 18 (26%) series of time-points where IRS was conducted, 41 (59%) had treated bed nets and 10 (14%) with MDA. The study dates ranged from 1931 to 2020 with 23 series of time-points from studies before 2000 (range ITN: 1993–2020, range MDA: 1931–2019, range IRS: 1961–2005).

The time-window of intervention implementation for clinical case data was mostly one month (55%), and never longer than a year. For prevalence, the implementation time-span was within a month for 63% of the series of time points, and for 98% within a year. The time-span for clinical case data collection was monthly data in 69% of the series of time points and for 97% of the studies it was less than a year. For patent infection data, the time-span of collection was a month or less for 65% of the series of time points and all but one had collection time windows of a year or less. [42]. The total number of patent infections found per time-point ranged from 1 to 789 cases.

#### **Risk of bias**

The majority of studies are before-after studies where the intervention is the same for all participants included at the time of implementation and the same participants are followed-up for both *P. vivax* and *P. falciparum* infection. Consequently, within a study we expect little bias arising from the selection of participants, classification of the intervention, or deviation from the intended intervention.

It is possible that the coverage of the intervention was not even throughout the study areas, and that some parts of the study areas may tend to have more *P. vivax* and some more *P. falcip-arum*. However, the same environmental characteristics tend to be conducive to both malaria species since they are transmitted by the same mosquito species, and the intention of the implementations were in most cases to provide the intervention to every resident.

There may be a bias in the proportion of infections that are patent by microscopy for the two species. *P. vivax* tends to have lower densities in the peripheral blood however the relative dynamics of sub-patent infections following an intervention in the two species are not well known.

There is likely to have been selection in which interventions were used in which settings. For example, MDA studies tended to be set in highly seasonal settings.

Since the primary aim of the studies was not to assess the relative impact, it is unlikely that selective reporting played a substantial role. A major reason for excluding studies was the lack of information on the intervention, particularly the timing, but it is not obvious that this would lead to bias in the relative impact.

#### Series of time-points with ITN

**Clinical Cases.** There were eleven studies [7,22,25,26,42–53] providing data on clinical malaria cases when a first-time distribution of ITNs took place, providing 26 series of time-points and 13 series of data points from six studies [7,22,43,45,46,51,54–56] in areas with repeated ITN distributions, where there had been a previous distribution or high coverage of bed nets (Fig 3).

	ITN (first distribution)	ITN (repeated distribution)	MDA (first round)	IRS (first spraying)	IRS (repeated spraying)	IRS (undetermined round)	Overall
Age group							
0-15 years	3 (20%)	2 (8%)	5 (50%)	1 (50%)	5 (100%)	10 (91%)	26 (38%)
2+ years	0 (0%)	1 (4%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	1 (1%)
5+ years	0 (0%)	0 (0%)	1 (10%)	0 (0%)	0 (0%)	0 (0%)	1 (1%)
15+ years	1 (7%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	1 (1%)
all ages	11 (73%)	23 (88%)	4 (40%)	1 (50%)	0 (0%)	1 (9%)	40 (58%)
Coverage <sup>1</sup>							
low	7 (47%)	4 (15%)	5 (50%)	0 (0%)	0 (0%)	0 (0%)	16 (23%)
high	7 (47%)	20 (77%)	4 (40%)	1 (50%)	5 (100%)	11 (100%)	48 (70%)
missing	1 (7%)	2 (8%)	1 (10%)	1 (50%)	0 (0%)	0 (0%)	5 (7%)
Seasonality <sup>2</sup>							
low	11 (73%)	24 (92%)	4 (40%)	2 (100%)	0 (0%)	8 (73%)	49 (71%)
high	4 (27%)	2 (8%)	6 (60%)	0 (0%)	5 (100%)	3 (27%)	20 (29%)
Relapse pattern							
frequent	12 (80%)	25 (96%)	6 (60%)	2 (100%)	0 (0%)	0 (0%)	45 (65%)
long	0 (0%)	0 (0%)	1 (10%)	0 (0%)	0 (0%)	0 (0%)	1 (1%)
both	3 (20%)	1 (4%)	3 (30%)	0 (0%)	5 (100%)	11 (100%)	26 (33%)
Transmission <sup>3</sup>							
low Pf low Pv	3 (20%)	12 (46%)	0 (0%)	0 (0%)	1 (20%)	5 (45%)	21 (30%)
low Pf high Pv	5 (33%)	0 (0%)	2 (20%)	0 (0%)	4 (80%)	6 (55%)	17 (25%)
high Pf low Pv	1 (7%)	12 (46%)	1 (10%)	0 (0%)	0 (0%)	0 (0%)	14 (20%)
high Pf high Pv	6 (40%)	2 (8%)	7 (70%)	2 (100%)	0 (0%)	0 (0%)	17 (25%)
Diagnostic tool							
microscopy	10 (67%)	12 (46%)	8 (80%)	2 (100%)	5 (100%)	11 (100%)	48 (70%)
PCR	3 (20%)	12 (46%)	2 (20%)	0 (0%)	0 (0%)	0 (0%)	17 (25%)
microscopy and PCR	2 (13%)	2 (8%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	4 (6%)
Type of study							
observational	12 (80%)	26 (100%)	10 (100%)	1 (50%)	5 (100%)	9 (82%)	63 (91%)
trial	3 (20%)	0 (0%)	0 (0%)	1 (50%)	0 (0%)	2 (18%)	6 (9%)
Survey taken in dif	ferent seasons <sup>4</sup>						
yes	6 (40%)	3 (12%)	9 (90%)	0 (0%)	0 (0%)	1 (9%)	19 (28%)
no	9 (60%)	2 (8%)	1 (10%)	1 (50%)	5 (100%)	10 (91%)	28 (41%)
unknown	0 (0%)	21 (81%)	0 (0%)	1 (50%)	0 (0%)	0 (0%)	22 (32%)
Time points per tir	ne series						
mean	2.8	2.3	5.6	3.5	2	2.1	2.9
median (min,max)	2 (2,6)	2 (2,5)	4.5 (2,13)	3.5 (2,5)	2 (2,2)	2 (2,3)	2(2,13)

#### Table 3. Characteristics of the 69 series of time-points of patent infections by intervention.

<sup>1</sup>High coverage is defined as more than 70% of people with access to intervention for ITN and 90% of people for IRS. For MDA high coverage is defined as more than 90% receiving at least one treatment

<sup>2</sup>Seasonality is defined using the measure of relative entropy described by Pascale (2015) [29] which quantifies the distribution of rainfall across the year

<sup>3</sup>Transmission intensity cut off was at 5% prevalence for the respective species or less than 100 cases per 1000 people per year

<sup>4</sup>The data collection may spread across different seasons or all the data points described may be collected throughout the same season (e.g. always during the wet season)

https://doi.org/10.1371/journal.pntd.0010541.t003

For the first-time distributions of ITNs, the proportion of cases that were caused by *P. vivax* tended to increase for most of the series (at least initially), but there were four (15% e), g), h), l)) that remained stable and five (19% a),d),i),m),y)) that decreased. Overall, there is substantial



**Fig 3. Proportion of clinical cases with** *P. vivax* **infection by time since first ITN distribution (white background) or repeated ITN distribution (grey background).** Blue filled circles: proportion of cases with *P. vivax* infection up to 24 months from the intervention. Open circles: over 24 months after the intervention happened (excluded in the analysis because focus was on the dynamics in the first 2 years after the intervention). Orange diamonds: the proportions in the control groups. Dots at time-point 0 were not necessarily measured at this time but at time-points before the intervention (median 2.2 months before implementation, range 0–24 months before) and it was assumed that this represents the proportion just before the intervention was implemented (Methods, Definitions of variables, Time since implemention). Dot size depends on number of cases found.

variability across the different settings. The study by Maude [50] included series in several provinces in Cambodia, where the proportions tended to increase for the first 24 months (Fig 3N-3V). Increases can also be seen in Papua New Guinea (Hetzel [42], and Ome-Kaius [7]). In contrast, a steep downward trend is visible in a series (Fig 3D) from Meghalaya in the North East of India. Some of the studies had other interventions in place (S2 Supporting Information) such as IRS. In some areas there were already nets, however, no previous distribution was mentioned and the coverage was not reported to have been high. Of the 13 series, that received a repeated distribution of ITNs six (46% aa),af),ag),ai),aj),al)) showed little or no change, five (38% ab),ac),ae),ah)am)) increased, whereas two (15%,ad),ak)) showed a downward trend (Fig 3). One series from Vanuatu shows a clear upward trend (Fig 3am)). In this case, the second distribution happened three years after the first one (the first one is shown in Fig 3Y). In the first distribution, the ITN coverage was around 25% whereas the second it was 50% of individuals [45]. Some studies had spraying or vector breeding site reduction in place [55,56,51] (S2 Supporting Information). The time difference between the previous and the

repeated distributions could vary (varied from approximately three to six years after previous distribution) which means nets from previous studies could have different remaining levels of protection.

**Patent infections.** For the first-time ITN distributions, a total of nine studies [5,7,22,25,26,42,43,52,53,57–60] provided information on patent infections. This led to 15 series of which twelve were observational before and after studies. Three series provide information on control groups, of which two also report patent infections before the intervention was implemented. The studies with a first-time distribution of ITNs were located in Papua New Guinea, Cambodia, Myanmar, Solomon Islands, Afghanistan and Pakistan. Eleven (73%) show a clear increase in the proportion of patent infections that were *P. vivax* (Fig 4) three studies (20%) show a decrease and one (7%) there was little change.

For the repeated distributions, there were 26 series of time-points from seven studies [5,7,22–24,43,61–66] which were mostly located in Papua New Guinea, with single studies in Cambodia, on the Thai-Myanmar border and in Ethiopia. Of the 26 series of time-points, twelve (46%) show



**Fig 4. Proportion of patent infections that are** *P. vivax* by time since first ITN distribution (white background) or repeated ITN distribution (grey background). Blue filled circles: proportion of cases attributed to *P. vivax* up to 24 months from the intervention. Orange diamonds: the proportions in the control groups. Dot size depending on number of infections found. Study p), r) and aa) are excluded from the regression analyses due to having no data collected in the first 24 months following the intervention. Pre-intervention points as described in Methods (definition of variables: time since implementation).

https://doi.org/10.1371/journal.pntd.0010541.g004

an upward trend, eight (31%) a downward trend and six (23%) show no obvious change (Fig 4). Fig 4K) and 4O) suggest that there are only slight differences in the patterns if the proportion of patent infections of the intervention group is compared to the control group instead of a before and after comparison. However, only two studies were available with both before and after data as well as a control group and this could not be investigated further.

#### Series of time-points with MDA

**Clinical cases.** There were five studies [52,53,67–70] with MDA providing 17 series of time-points, of which 13 were first-time administrations and four had received MDA three to six months previously. The studies were located in Brazil, Nicaragua, India and Myanmar. All studies were observational. Upward trends are observed in two series (12%, k) and p)) (Fig 5). For the remaining studies, the observed trends are either non-linear or unclear (41%, a), b),c), e), f), g) and l)).

Different drugs were used in the studies. Some used primaquine (S2 Supporting Information), however, the doses were not sufficient to kill hypnozoites and prevent relapses. The recommendation by the WHO to prevent relapses is to treat people for 14 days with a dose of 0.25–0.5mg/kg per day [71]. Therefore with the drug regimens used only blood stage parasites were cleared. The prophylactic period of the drugs varied. Some studies gave drugs on several occasions over three months. Most of the studies had other interventions in place in these areas (S2 Supporting Information).

**Patent infections.** There were ten series of time-points from six studies [52,53,68,72–76] providing data on MDA and patent infections. All the series report on first time distributions of MDA and were observational. They were located in Brazil, Papua New Guinea, Myanmar, Indonesia and Palestine. There were eight studies with an initial upward and two studies with initial downward observed trends (Fig 6). The only study with sufficient primaquine to prevent



**Fig 5. Proportion of clinical cases with** *P. vivax* **infection by time since MDA**. Blue dots: first round of MDA, red dots: repeated MDA (first round took place 3–6 months before). Open circles: more than 3 months after MDA. Dot size depending on number of cases found Pre-intervention points as described in Methods (definition of variables: time since implementation).

https://doi.org/10.1371/journal.pntd.0010541.g005



Fig 6. Proportion of patent infections that are *P. vivax* by time since MDA. Unfilled dots: more than 3 months after mass drug administration. Dot size depending on number of infections found Pre-intervention points as described in Methods (definition of variables: time since implementation).

relapses was by Hofmann [72] (Fig 6C)) with treatment with primaquine for twenty days with a dose of 0.5mg/kg/day in children aged 5 to 10 years. The study conducted by Kliger [74] treated with plasmochine and quinine sulphate but the dose was lower than that needed to kill the hypnozoites in these settings [77]. The study by Van Dijk [75] did 11 rounds of mass drug administration measuring patent infections before and after 10 rounds.

#### Series of time-points with IRS

**Clinical cases.** There was a total of 16 series of time-points with information on any round of IRS from eleven different studies [40,47,56,78–85]. The studies were located in Venezuela, Ethiopia, India, Vietnam, Brazil and Pakistan. All apart from two were observational studies. Five series show at least slight initial downward patterns (31%, a), d), e), j), k)) (Fig 7). Upward patterns can be observed in three studies (19%, i) o),p)). The remaining eight studies (50%) show little change or have a large amount of stochasticity. For four series, there had been no previous IRS. For nine series, there was no spraying mentioned previous to the study however, it was also not clearly indicated that there was none. In the study by Doke et al [81], there was previous spraying with DDT but due to resistance this was changed to lambdacyha-lothrin. Various insecticides were used in the studies (S2 Supporting Information).

**Patent infections.** There were 18 series of time-points from six studies [60,84–88], of which 16 were from observational studies and two had control groups. The studies were located in Indonesia, Pakistan, the Solomon Islands and India. For two series it was the first spraying, for five it was likely that spraying had happened before, and for the remaining studies it was unclear. There were nine series of time-points that showed a clear upward trend in the proportion of cases attributed to *P. vivax* (Fig.8). Five series of time-points show downward trends and the remaining four studies show little (initial) change.

# Graphical summaries of the proportion of cases or infections that were *P*. *vivax* over time by factor

Overall, there was substantial variability in the patterns over time. We used spaghetti plots to investigate visually if there are general patterns by each factor. Other characteristics of the series are described in S1.



Fig 7. Proportion of clinical cases with *P. vivax* infection by time since IRS. Blue: first time spraying (there was no spraying for at least a year before), red: repeated spraying (spraying had occurred previously but there was a change in insecticide used), green: unclear if spraying happened before (no mention of spraying before the study but also no clear mentioning of spraying being the first time). Open circles: more than 24 months after IRS, Orange diamond shaped dots are from the control group. Dot size depending on number of cases. Pre-intervention points as described in Methods (definition of variables: time since implementation).



**Fig 8. Proportion of patent infections that are** *P. vivax* **by time since IRS.** Blue: first time spraying (there was no spraying for at least a year before), red: repeated spraying (spraying happened before but there was a change in insecticide used in panel f) and g)), green: unclear if spraying happened before (no mentioning of spraying before the study but also no clear mentioning of spraying being the first time). Open circles: more than 24 months after IRS, Orange diamond shaped dots are from the control group. Dot size depending on number of cases found. Pre-intervention points as described in Methods (definition of variables: time since implementation).

https://doi.org/10.1371/journal.pntd.0010541.g008



Fig 9. Spaghetti plots splitting the series of time-points of the proportion of cases with *P. vivax* infection by relapse pattern, coverage, seasonality, initial proportion of *P. vivax* and transmission intensity after first time ITN distribution. Dot size depending on number of infections.

**Series with ITN by factor.** With first time distribution of ITNs with clinical cases, there were different trends observed over time by relapse patterns with a stronger tendency for upward trends in the frequent relapse category (Fig 9). The same is true for high coverage, however, many of the studies located in frequent relapse pattern areas also received interventions with high coverage. No clear differences were observed for the other factors. Age and transmission intensity could not be investigated since all but one studies locked at all ages apart from Ome-Kaius [7] (Fig 3W) with under 5-year-olds. Most studies were classified as having low transmission intensity for both *P. falciparum* and *P. vivax* with prevalence rates lower than 5% (or if prevalence was not available incidence lower than 100 cases per 1000 people per year was considered low).

For the repeated distribution of ITN with clinical cases, no clear patterns were observed for any of the factors (Fig 10). Some factors could not be assessed due to the data available.

To consider the hypothesis that the short-term dynamics may differ following first time and repeated ITN distributions, they were presented directly in the same graph. There is substantial variation in both groups (Fig 11), with a potential tendency for a slightly greater overall increase in the proportion that were *P. vivax* for the first time ITN distributions.



Fig 10. Spaghetti plots splitting the series of time-points of the proportion of cases with *P. vivax* infection by relapse pattern, transmission intensity and coverage after repeated ITN distribution. Excluding series of time points that have no measurement within 24 months after the intervention. Dot size depending on number of infections.

For a first-time distribution of ITN with patent infections, visually the variables with the most distinct patterns between categories were relapse patterns and coverage (Fig 12). When there was a repeated distribution with patent infections, none of the factors had obvious differences between categories of any factors. Relapse pattern, seasonality and transmission intensity could not be investigated due to data availability (Fig 13).



Fig 11. Spaghetti plots splitting the series of time-points of the proportion of cases with *P. vivax* infection by round of ITN distribution. Excluding series of time points that have no measurement within 24 months after the intervention. Dot size depending on number of infections.

https://doi.org/10.1371/journal.pntd.0010541.g011



Fig 12. Spaghetti plots splitting the series of time-points of the proportion of patent infections that are *P. vivax* by relapse pattern, coverage, transmission intensity, seasonality and initial proportion of *P. vivax* after first time distribution of ITNs. Dot size depending on number of infections.

When plotted by distribution, there was no clear difference in the patterns over time by round of ITN distribution (Fig 14).

**Series with mass drug administration by factor.** The first and repeated rounds of MDA were not separated for the analysis because MDA is assumed to act most strongly in the short-term and therefore similar effects were expected for first and repeated rounds of MDA.

**Clinical cases:** Most of the spaghetti plots (Fig 15) show no clear patterns in the proportion of clinical cases that were *P. vivax* by time and there was substantial variation in all categories. The studies with three rounds of mass drug administration tended to have longer follow-up times. Coverage was considered high for all studies where 90% or more of people received at least one round of MDA. There appears to be an upward trend in the proportion of cases attributed to *P. vivax* with high coverage, for studies receiving three rounds of MDA, frequent relapse pattern areas as well as low *P. falciparum* and high *P. vivax* transmission. However, it is mostly the same studies for the categories for all of these variables. The time-period of the study may also be a potential factor but there were only three studies later than the year 2000



Fig 13. Spaghetti plots splitting the series of time-points of the proportion of patent infections that are *P. vivax* by initial proportion of *P. vivax*, coverage and transmission intensity after repeated distribution of ITNs. Excluding series of time points that have no measurement within 24 months after the intervention. Dot size depending on number of infections.

and these tended to have higher coverage, three rounds instead of one, and longer follow-up periods. Migration from areas with a high malaria transmission intensity was not captured in the data but was given as one possible explanation for the studies in India by Kondrashin et al. [69] for the areas where *P. vivax* decreases but *P. falciparum* does not.



**Fig 14.** Spaghetti plots splitting the series of time-points of the proportion of patent infections that are *P. vivax* by round of ITN distribution. Excluding series of time points that have no measurement within 24 months after the intervention. Dot size depending on number of infections.

**Patent infections:** Overall there were no clear patterns in the proportion of cases with *P*. *vivax* parasitemia (Fig 16) apart from series in long latency relapse pattern areas where there appears to be an increase. Other factors such as rounds of MDA were not considered as all of the studies had three rounds of MDA or more in total every month apart from Hofmann *et al* [72] with one round over twenty days and McGreevy *et al* [68] where the number of rounds was not recorded.

**Series with indoor residual spraying by factor.** For indoor residual spraying, for both clinical cases (Fig 17) and patent infections (Fig 18), no clear visual patterns for any of the variables emerged. There was a substantial amount of variation.

**Regression analyses to estimate the observed trends in the proportion of** *P. vivax* cases **by time since intervention and identify potential factors.** To estimate the changes over time since the intervention and to identify factors that are associated with the change over time, logistic regression was used with random effects for the intercept and slope for series (S3 Supporting Information). When possible, models included several explanatory variables, however, since the number of series for each level of the covariates is limited and an interaction term for the effect of the covariates is needed to estimate the effect over time the full models did not always converge. In those cases, the analysis was carried out for one factor at a time.

The analyses were done separately for the interventions as the effects of the explanatory variables might differ by intervention. Adjustment for the season at the time of the surveys was included in all models.

In summary, for all interventions there is substantial variability in the trends of the proportion of clinical cases or patent infections that were *P. vivax* in the short-term by time since the implementation (Tables 4 & 5). The time-points used were up to 24 months after the intervention for ITN and IRS and up to three months for MDA. Within these time periods, the patterns were roughly linear.

When the interventions are compared overall, there is a tendency for upward patterns in malaria attributed to *P. vivax* for ITN and IRS, with MDA estimates uncertain in the first three months, although these associations are not significant. There was no evidence that the round



Fig 15. Spaghetti plots splitting the series of time-points of the proportion of cases with *P. vivax* infection by rounds of MDA, coverage, relapse pattern and transmission intensity after MDA. Dot size depending on number of infections.

of intervention for ITN differs significantly in a model with time however the effects of the explanatory variables differ by round of distribution. Overall, the analyses were unable to rule in or rule out explanatory factors due to the limited number of series and the collinearity of some variables. There was evidence of an effect of coverage for some of the interventions, however not all of them and this could not be investigated for IRS. Transmission intensity had significantly different effects by species for ITN (Fig 19) but no clear pattern was observed for MDA and IRS. The results suggest that relapse pattern may be able to explain some of the variation with long latency patterns tending to lead to a weaker increase in *P. vivax* for ITN as well as for IRS. There is uncertainty in case of seasonality as it could not be tested for all interventions but possibly had an effect in others. The initial proportion of *P. vivax* is a possible factor, however, this is uncertain and it was not significant in most analyses.

#### Discussion

To our knowledge, this is the first systematic review of the relative effect of interventions on sympatric *P. vivax* and *P. falciparum*. The mean increase in the proportion of cases that were



Fig 16. Spaghetti plots splitting the series of time-points of the proportion of patent infections that are *P. vivax* by relapse pattern, coverage, transmission intensity, initial proportion of *P. vivax* and seasonality after MDA. Dot size depending on number of infections.

*P. vivax* either significantly increased or the estimates were in the direction of an increase following ITN, IRS and for MDA after the first three months. However, the most striking finding was the substantial variability found in the relative impact for all of the interventions over the time-scale of this study: up to 24 months for ITN and IRS and up to six months for MDA.

A recent review by Price [6], found that the proportion of *P. vivax* tends to increase when malaria control interventions are implemented, over much longer time scales.

Due to the number of series, the variability in the observed trends and collinearity between variables, it was not possible to rule in or rule out many of the variables as having an impact on the trend over time. Variables which are likely to have an effect were transmission intensity, coverage and relapse pattern. Transmission intensity was either significantly associated or had large odds ratios in several, but not all, of the analyses. It is likely that transmission intensity plays a role as the effect of interventions can be different for single species for different transmission intensities and different transmission levels can pose different challenges such as more heterogeneity and low level parasitemia in low transmission areas [89].

Even though there was no evidence for an impact of coverage for all interventions, it may have an influence if an intervention does have different relative effects. High coverage may



Fig 17. Spaghetti plots splitting the series of time-points of the proportion of cases with *P. vivax* infection by seasonality, initial proportion of *P. vivax* and relapse pattern after IRS. Dot size depending on number of infections.



Fig 18. Spaghetti plots splitting the series of time-points of the proportion of patent infections that are *P. vivax* by seasonality, initial proportion of *P. vivax* and transmission intensity after IRS. Dot size depending on number of infections.

https://doi.org/10.1371/journal.pntd.0010541.g018

Intervention	Time-span	Number of series of time points	Change over time: Odds ratio per month	95% CI	Variation between series (min/max estimated odds ratio per month)
ITN (first round)	24 months	26	1.03	1.01-1.06	0.92-1.13
ITN (repeated round)	24 months	12	1.04	1.01-1.07	0.99–1.10
MDA	<3 months	17	0.93	0.63-1.39	0.37-2.39
MDA	3–6 months	13	1.62	1.14-2.30	1.08-2.06
IRS	24 months	16	1.00	0.95-1.06	0.87-1.24

Table 4. Estimated changes in the proportion of clinical cases that were *P. vivax* over time for each intervention separately, adjusted for season at time-point of survey.

https://doi.org/10.1371/journal.pntd.0010541.t004

also potentially lead to a stronger reduction in transmission levels which in turn could affect levels of acquired immunity. Relapse pattern may have an effect with differences between long latency and frequent relapse pattern. The long latency relapse pattern, which has fewer relapses [27], might lead to weaker increases in the proportion of cases attributed to *P. vivax* (as was found with significance for first time ITN distribution patent infections and IRS clinical cases, and in the direction of a weaker increase though not significant for the first three months after MDA clinical cases and first time ITN distribution clinical cases). The effects of seasonality and the initial proportion of *P. vivax* were uncertain.

Other factors could be important but could not be investigated due to the limited data. There were some variables for which data was collected but there was little variety in values such as age and diagnostic test. For some of the considered factors more detailed information would have been beneficial such as previous coverage of the intervention or inequality in coverage across a community. Although the season at which data points were collected was considered, the exact timing during the season may be informative [90]. There were also some variables for which data was not collected but could have an impact, for example, drug resistance which could vary for the two species. Other factors could be migration, access to treatment, recent changes in transmission or previous interventions. The extent of competition and cross-immunity between the species is uncertain.

The results can provide information for planning and evaluating the impact of interventions. It is important to consider the biological differences between *P. vivax* and *P. falciparum* when evaluating an intervention which does not target hypnozoites. The species should be monitored separately. This would allow the adjustment of the strategy if the original intervention is less successful for one species. To aid interpretation of the observed patterns, contextual factors that could influence the malaria situation and changes in the species composition should be monitored. These factors include decreasing coverage of other interventions, human behavior such as migration, or emergence of drug resistance. There can also be changes in detection (more access to health facilities, changes in diagnostic tools) or changes in reporting (changes in case definitions, biases in selection of survey populations) [91].

More clarity could be obtained by including more studies in the analysis if they become available. However, as transmission decreases further, it may be more appropriate to employ

Table 5.	Estimated changes in the proportion of patent infections that were P. vivax over time for each intervention separately, adjusted for season at time-point o
survey.	

Intervention	Time-span	Number of series of time points	Change over time: Odds ratio per month	95% CI	Variation between series (min/max)
ITN	24 months	37	1.02	0.98-1.06	0.82-1.15
MDA	<3 months	9	0.97	0.57-1.65	0.37-1.83
IRS	24 months	18	1.27	1.08-1.51	0.93–1.75

https://doi.org/10.1371/journal.pntd.0010541.t005





targeted or reactive interventions. It would be useful if studies describe interventions in detail, information was often lacking on the timing of the intervention as well as coverage. There were also studies that reported that there was a national control program but did not specify what this entailed and therefore could not be included in the analysis.

Where data is limited, a possible further step to gain clarity is to use modelling. Predictions can be made of the effect of variables for which there was limited data and validated models can be used to disentangle the effects of collinear covariates.

#### Conclusion

When interventions are implemented in areas with sympatric *P. vivax* and *P. falciparum* malaria, there is substantial variation in how the proportions of the two species change on a short-term time scale, with a tendency for the proportion of *P. vivax* to increase overall. It was not possible to rule in or out many of the potential factors, due to the high variability and the availability of data. Relapse pattern, coverage, transmission and intervention type are possible explanations contributing to this variation but they were not consistently significant. More studies with well-described intervention implementations would reduce the uncertainty.

#### **Supporting information**

**S1** Supporting Information. Included studies and variables by series of time-points. (DOCX)

**S2 Supporting Information. Interventions in place at baseline.** (DOCX)

**S3 Supporting Information. Results from the regression analyses.** (DOCX)

#### Acknowledgments

We thank Jacqueline Huber for helping with the development of the literature search strategy, and everyone within Swiss TPH who provided insights and feedback especially Manuel Hetzel, Guojing Yang and Christian Lengeler.

#### **Author Contributions**

Conceptualization: Melanie Loeffel, Amanda Ross.

Data curation: Melanie Loeffel.

Formal analysis: Melanie Loeffel, Amanda Ross.

Methodology: Melanie Loeffel, Amanda Ross.

Writing - original draft: Melanie Loeffel.

Writing - review & editing: Melanie Loeffel, Amanda Ross.

#### References

- 1. World Health Organization. World malaria report 2021. Geneva: World Health Organization; 2021.
- Battle KE, Lucas TCD, Nguyen M, Howes RE, Nandi AK, Twohig KA et al. Mapping the global endemicity and clinical burden of *Plasmodium vivax*, 2000–17: a spatial and temporal modelling study. *Lancet*. 2019; 394(10195):332–43. https://doi.org/10.1016/S0140-6736(19)31096-7 PMID: 31229233
- Weiss DJ, Lucas TCD, Nguyen M, Nandi AK, Bisanzio D, Battle KE et al. Mapping the global prevalence, incidence, and mortality of *Plasmodium falciparum*, 2000–17: a spatial and temporal modelling study. *Lancet*. 2019; 394(10195):322–31. https://doi.org/10.1016/S0140-6736(19)31097-9 PMID: 31229234
- Phimpraphi W, Paul RE, Yimsamran S, Puangsa-art S, Thanyavanich N, Maneeboonyang W et al. Longitudinal study of *Plasmodium falciparum* and *Plasmodium vivax* in a Karen population in Thailand. *Malar J* 2008; 7:99. https://doi.org/10.1186/1475-2875-7-99 PMID: 18518964
- Koepfli C, Ome-Kaius M, Jally S, Malau E, Maripal S, Ginny J et al. Sustained malaria control over an 8-Year period in Papua New Guinea: The challenge of low-density asymptomatic Plasmodium Infections. *J Inf Dis* 2017; 216(11):1434–43. https://doi.org/10.1093/infdis/jix507 PMID: 29029179
- Price RN, Commons RJ, Battle KE, Thriemer K, Mendis K. *Plasmodium vivax* in the Era of the Shrinking *P. falciparum* Map. *Trends Parasitol* 2020; 36(6):560–70. https://doi.org/10.1016/j.pt.2020.03.009
  PMID: 32407682
- Ome-Kaius M, Kattenberg JH, Zaloumis S, Siba M, Kiniboro B, Jally S et al. Differential impact of malaria control interventions on *P. falciparum* and *P. vivax* infections in young Papua New Guinean children. BMC Med. 2019; 17(1):220. https://doi.org/10.1186/s12916-019-1456-9 PMID: 31813381
- 8. Chu CS, Carrara VI, Parker DM, Proux S, Charunwatthana P, McGready R et al. Declining Burden of *Plasmodium vivax* in a population in Northwestern Thailand from 1995 to 2016 before comprehensive primaquine prescription for radical cure. Am J Trop Med Hyg. 2020; 102(1):147–50.
- Phillips MA, Burrows JN, Manyando C, van Huijsduijnen RH, Van Voorhis WC, Wells TNC. Malaria. Nature Reviews Disease Primers 2017; 3:17050. https://doi.org/10.1038/nrdp.2017.50 PMID: 28770814

- Briët OJ, Penny MA. Repeated mass distributions and continuous distribution of long-lasting insecticidal nets: modelling sustainability of health benefits from mosquito nets, depending on case management. *Malar J.* 2013; 12:401. https://doi.org/10.1186/1475-2875-12-401 PMID: 24200296
- Poirot E, Skarbinski J, Sinclair D, Kachur SP, Slutsker L, Hwang J. Mass drug administration for malaria. *Cochrane Database of Systematic Reviews* 2013; 2013(12):CD008846. https://doi.org/10. 1002/14651858.CD008846.pub2 PMID: 24318836
- Pluess B, Tanser FC, Lengeler C, Sharp BL. Indoor residual spraying for preventing malaria. The Cochrane Database of Systematic Reviews. 2010; 2010(4):CD006657. <u>https://doi.org/10.1002/ 14651858.CD006657.pub2</u> PMID: 20393950
- Pryce J, Richardson M, Lengeler C. Insecticide-treated nets for preventing malaria. Cochrane Database of Systematic Reviews. 2018; 11(11):CD000363. https://doi.org/10.1002/14651858.CD000363.pub3 PMID: 30398672
- Lengeler C. Insecticide-treated bed nets and curtains for preventing malaria. Cochrane Database of Systematic Reviews. 2004(2):CD000363. https://doi.org/10.1002/14651858.CD000363.pub2 PMID: 15106149
- Choi L, Pryce J, Garner P. Indoor residual spraying for preventing malaria in communities using insecticide-treated nets. Cochrane Database of Systematic Reviews. 2019; 5(5):CD012688. https://doi.org/ 10.1002/14651858.CD012688.pub2 PMID: 31120132
- 16. Rohatgi A. WebPlotDigitizer [Internet]. 2021 [Available from: https://automeris.io/WebPlotDigitizer.
- White NJ. The assessment of antimalarial drug efficacy. *Trends Parasitol* 2002; 18(10):458–64. <u>https://doi.org/10.1016/s1471-4922(02)02373-5 PMID: 12377597</u>
- Ghani AC, Sutherland CJ, Riley EM, Drakeley CJ, Griffin JT, Gosling RD et al. Loss of population levels of immunity to malaria as a result of exposure-reducing interventions: consequences for interpretation of disease trends. *PLOS ONE* 2009; 4(2):e4383. <u>https://doi.org/10.1371/journal.pone.0004383</u> PMID: 19198649
- Gao B, Saralamba S, Lubell Y, White LJ, Dondorp AM, Aguas R. Determinants of MDA impact and designing MDAs towards malaria elimination. *eLife* 2020;9. https://doi.org/10.7554/eLife.51773 PMID: 32293559
- 20. Maitland K, Williams TN, Bennett S, Newbold CI, Peto TE, Viji J et al. The interaction between *Plasmodium falciparum* and *P. vivax* in children on Espiritu Santo island, Vanuatu. *Trans Roy Soc Trop Med Hyg* 1996; 90(6):614–20. https://doi.org/10.1016/s0035-9203(96)90406-x PMID: 9015495
- 21. Ross A, Koepfli C, Schoepflin S, Timinao L, Siba P, Smith T et al. The incidence and differential seasonal patterns of *Plasmodium vivax* primary infections and relapses in a cohort of children in Papua New Guinea. *PLOS Negl Trop Dis*2016; 10(5):e0004582.
- 22. Lin E, Kiniboro B, Gray L, Dobbie S, Robinson L, Laumaea A et al. Differential patterns of infection and disease with *P. falciparum* and *P. vivax* in young Papua New Guinean children. *PLOS ONE*. 2010; 5(2): e9047. https://doi.org/10.1371/journal.pone.0009047 PMID: 20140220
- 23. Deressa W, Yihdego YY, Kebede Z, Batisso E, Tekalegne A, Dagne GA. Effect of combining mosquito repellent and insecticide treated net on malaria prevalence in Southern Ethiopia: a cluster-randomised trial. *Parasit Vector* 2014; 7:132. https://doi.org/10.1186/1756-3305-7-132 PMID: 24678612
- 24. Luxemburger C, Perea WA, Delmas G, Pruja C, Pecoul B, Moren A. Permethrin-impregnated bed nets for the prevention of malaria in schoolchildren on the Thai-Burmese border. *Trans Roy Soc Trop Med Hyg* 1994; 88(2):155–9. https://doi.org/10.1016/0035-9203(94)90273-9 PMID: 8036656
- Rowland M, Bouma M, Ducornez D, Durrani N, Rozendaal J, Schapira A et al. Pyrethroid-impregnated bed nets for personal protection against malaria for Afghan refugees. *Trans Roy Soc Trop Med Hyg.* 1996; 90(4):357–61. https://doi.org/10.1016/s0035-9203(96)90505-2 PMID: 8882175
- Rowland M, Hewitt S, Durrani N, Saleh P, Bouma M, Sondorp E. Sustainability of pyrethroid-impregnated bednets for malaria control in Afghan communities. *Bull WHO*. 1997; 75(1):23–9. PMID: 9141747
- White NJ. Determinants of relapse periodicity in *Plasmodium vivax* malaria. *Malar J.* 2011; 10:297. https://doi.org/10.1186/1475-2875-10-297 PMID: 21989376
- Reiner RC Jr., Geary M, Atkinson PM, Smith DL, Gething PW. Seasonality of *Plasmodium falciparum* transmission: a systematic review. *Malar J*. 2015; 14:343. https://doi.org/10.1186/s12936-015-0849-2 PMID: 26370142
- Pascale S, Lucarini V, Feng X, Porporato A, ul Hasson S. Analysis of rainfall seasonality from observations and climate models. *Climate Dynamics* 2015; 44(11–12):3281–301.
- 30. Sterne JAC, Hernan MA, McAleenan A, Reeves BC, Higgins JPT. Chapter 25: Assessing risk of bias in a non-randomized study. In: Higgins JPT, Thomas J, Chandler J, Cumpston M, Li T, Page MJ, Welch VA (editors). Cochrane Handbook for Systematic Reviews of Interventions version 6.2 (updated February 2021).

- **31.** R Core Team. R: a language and environment for statistical computing. Vienna, Austria: R Foundation for Statistical Computing; 2020.
- 32. Iannone R. DiagrammeR: Graph/Network Visualization. 2020.
- Original S code by Richard A. Becker and Allan R. Wilks. R version by Ray Brownrigg. Enhancements by Thomas P Minka and Alex Deckmyn. maps: Draw Geographical Maps. 2018.
- 34. Wickham H. ggplot2: Elegant Graphics for Data Analysis: Springer-Verlag New York; 2016.
- 35. South A. rnaturalearth: World Map Data from Natural Earth. 2017.
- **36.** Wickham H. The Split-Apply-Combine Strategy for Data Analysis. Journal of Statistical Software. 2011; 40(1):1–29.
- Wickham H. Reshaping Data with the reshape Package. Journal of Statistical Software. 2007; 21 (12):1–20.
- Bates D, Mächler M, Bolker B, Walker S. Fitting linear mixed-effects models using Ime4. J Stat Software. 2015; 67(1):1–48.
- Hanafi-Bojd AA, Vatandoost H, Oshaghi MA, Haghdoost AA, Shahi M, Sedaghat MM et al. Entomological and epidemiological attributes for malaria transmission and implementation of vector control in southern Iran. Acta Trop 2012; 121(2):85–92. https://doi.org/10.1016/j.actatropica.2011.04.017 PMID: 21570940
- 40. Kattenberg JH, Erhart A, Truong MH, Rovira-Vallbona E, Vu KAD, Nguyen THN et al. Characterization of *Plasmodium falciparum* and *Plasmodium vivax* recent exposure in an area of significantly decreased transmission intensity in Central Vietnam. *Malar J* 2018; 17(1):180. <u>https://doi.org/10.1186/s12936-018-2326-1</u> PMID: 29703200
- Graves PM, Brabin BJ, Charlwood JD, Burkot TR, Cattani JA, Ginny M et al. Reduction in incidence and prevalence of *Plasmodium falciparum* in under-5-year-old children by permethrin impregnation of mosquito nets. *Bull WHO*. 1987; 65(6):869–77. PMID: 3325185
- 42. Hetzel MW, Reimer LJ, Gideon G, Koimbu G, Barnadas C, Makita L et al. Changes in malaria burden and transmission in sentinel sites after the roll-out of long-lasting insecticidal nets in Papua New Guinea. *Parasit Vector* 2016; 9(1):340. https://doi.org/10.1186/s13071-016-1635-x PMID: 27301964
- **43.** Betuela I, Rosanas-Urgell A, Kiniboro B, Stanisic DI, Samol L, de Lazzari E et al. Relapses contribute significantly to the risk of *Plasmodium vivax* infection and disease in Papua New Guinean children 1–5 years of age. J Inf Dis 2012; 206(11):1771–80.
- 44. Kessler A, van Eijk AM, Jamir L, Walton C, Carlton JM, Albert S. Malaria in Meghalaya: a systematic literature review and analysis of data from the National Vector-Borne Disease Control Programme. *Malar J* 2018; 17(1):411. https://doi.org/10.1186/s12936-018-2563-3 PMID: 30400879
- Chaves LF, Kaneko A, Taleo G, Pascual M, Wilson ML. Malaria transmission pattern resilience to climatic variability is mediated by insecticide-treated nets. *Malar J* 2008; 7:100. <u>https://doi.org/10.1186/</u> 1475-2875-7-100 PMID: 18518983
- 46. Kaneko A, Taleo G, Kalkoa M, Yaviong J, Reeve PA, Ganczakowski M et al. Malaria epidemiology, glucose 6-phosphate dehydrogenase deficiency and human settlement in the Vanuatu Archipelago. Acta Trop 1998; 70(3):285–302. https://doi.org/10.1016/s0001-706x(98)00035-7 PMID: 9777715
- Loha E, Deressa W, Gari T, Balkew M, Kenea O, Solomon T et al. Long-lasting insecticidal nets and indoor residual spraying may not be sufficient to eliminate malaria in a low malaria incidence area: results from a cluster randomized controlled trial in Ethiopia. *Malar J*. 2019; 18(1):141. <u>https://doi.org/ 10.1186/s12936-019-2775-1</u> PMID: 30999957
- 48. Shah NK, Tyagi P, Sharma SK. The impact of artemisinin combination therapy and long-lasting insecticidal nets on forest malaria incidence in tribal villages of India, 2006–2011. PLOS ONE 2013; 8(2): e56740. https://doi.org/10.1371/journal.pone.0056740 PMID: 23437229
- Sahu SS, Jambulingam P, Vijayakumar T, Subramanian S, Kalyanasundaram M. Impact of alphacypermethrin treated bed nets on malaria in villages of Malkangiri district, Orissa, India. Acta Tropica 2003; 89(1):55–66. https://doi.org/10.1016/j.actatropica.2003.07.002 PMID: 14636983
- Maude RJ, Nguon C, Ly P, Bunkea T, Ngor P, Canavati de la Torre SE et al. Spatial and temporal epidemiology of clinical malaria in Cambodia 2004–2013. *Malar J* 2014; 13:385. <u>https://doi.org/10.1186/</u> 1475-2875-13-385 PMID: 25266007
- Sena LD, Deressa WA, Ali AA. Analysis of trend of malaria prevalence in south-west Ethiopia: a retrospective comparative study. *Malar J.* 2014; 13:188. <u>https://doi.org/10.1186/1475-2875-13-188</u> PMID: 24886236
- Chaumeau V, Kajeechiwa L, Fustec B, Landier J, Naw Nyo S, Nay Hsel S et al. Contribution of asymptomatic Plasmodium infections to the transmission of malaria in Kayin State, Myanmar. J Inf Dis. 2019; 219(9):1499–509.

- 53. Landier J, Kajeechiwa L, Thwin MM, Parker DM, Chaumeau V, Wiladphaingern J et al. Safety and effectiveness of mass drug administration to accelerate elimination of artemisinin-resistant falciparum malaria: A pilot trial in four villages of Eastern Myanmar. *Wellcome Open Res* 2017; 2:81. <u>https://doi.org/10.12688/wellcomeopenres.12240.1</u> PMID: 29062913
- Rodriguez-Rodriguez D, Maraga S, Lorry L, Robinson LJ, Siba PM, Mueller I et al. Repeated mosquito net distributions, improved treatment, and trends in malaria cases in sentinel health facilities in Papua New Guinea. *Malar J* 2019; 18(1):364. https://doi.org/10.1186/s12936-019-2993-6 PMID: 31718659
- 55. Mitjà O, Paru R, Selve B, Betuela I, Siba P, De Lazzari E et al. Malaria epidemiology in Lihir Island, Papua New Guinea. *Malar J*. 2013; 12:98. https://doi.org/10.1186/1475-2875-12-98 PMID: 23497296
- 56. Loha E, Lunde TM, Lindtjørn B. Effect of bednets and indoor residual spraying on spatio-temporal clustering of malaria in a village in south Ethiopia: a longitudinal study. PLOS ONE. 2012; 7(10):e47354. https://doi.org/10.1371/journal.pone.0047354 PMID: 23077598
- Thomson R, Sochea P, Sarath M, MacDonald A, Pratt A, Poyer S et al. Rubber plantations and drug resistant malaria: a cross-sectional survey in Cambodia. Malar J. 2019; 18(1):379. <u>https://doi.org/10. 1186/s12936-019-3000-y PMID: 31775755</u>
- 58. Rowland M, Webster J, Saleh P, Chandramohan D, Freeman T, Pearcy B et al. Prevention of malaria in Afghanistan through social marketing of insecticide-treated nets: evaluation of coverage and effectiveness by cross-sectional surveys and passive surveillance. *Trop Med Int Health* 2002; 7(10):813–22. https://doi.org/10.1046/j.1365-3156.2002.00940.x PMID: 12358615
- Smithuis FM, Kyaw MK, Phe UO, van der Broek I, Katterman N, Rogers C et al. The effect of insecticide-treated bed nets on the incidence and prevalence of malaria in children in an area of unstable seasonal transmission in western Myanmar. *Malar J*. 2013; 12:363. https://doi.org/10.1186/1475-2875-12-363 PMID: 24119916
- Hii JL, Kanai L, Foligela A, Kan SK, Burkot TR, Wirtz RA. Impact of permethrin-impregnated mosquito nets compared with DDT house-spraying against malaria transmission by Anopheles farauti and An. punctulatus in the Solomon Islands. *Med Vet Entomol.* 1993; 7(4):333–8. <u>https://doi.org/10.1111/j. 1365-2915.1993.tb00701.x PMID: 8268487</u>
- Hetzel MW, Pulford J, Ura Y, Jamea-Maiasa S, Tandrapah A, Tarongka N et al. Insecticide-treated nets and malaria prevalence, Papua New Guinea, 2008–2014. *Bull WHO*. 2017; 95(10):695–705b. <u>https://</u> doi.org/10.2471/BLT.16.189902 PMID: 29147042
- Hetzel MW, Morris H, Tarongka N, Barnadas C, Pulford J, Makita L et al. Prevalence of malaria across Papua New Guinea after initial roll-out of insecticide-treated mosquito nets. Trop Med Int Health. 2015; 20(12):1745–55. https://doi.org/10.1111/tmi.12616 PMID: 26427024
- Hetzel MW, Saweri OP, Kuadima JJ, Smith I, Ura Y, Tandrapah A et al. Papua New Guinea malaria indicator survey 2016–2017: malaria prevention, infection and treatment. Papua New Guinea Institue of Medical Research, Goroka, 2018; 2018.
- 64. Sluydts V, Heng S, Coosemans M, Van Roey K, Gryseels C, Canier L et al. Spatial clustering and risk factors of malaria infections in Ratanakiri Province, Cambodia. Malar J 2014; 13:387. <u>https://doi.org/10.1186/1475-2875-13-387</u> PMID: 25269827
- 65. Sluydts V, Durnez L, Heng S, Gryseels C, Canier L, Kim S et al. Efficacy of topical mosquito repellent (picaridin) plus long-lasting insecticidal nets versus long-lasting insecticidal nets alone for control of malaria: a cluster randomised controlled trial. Lancet Inf Dis. 2016; 16(10):1169–77. https://doi.org/10. 1016/S1473-3099(16)30148-7 PMID: 27371977
- Kattenberg JH, Gumal DL, Ome-Kaius M, Kiniboro B, Philip M, Jally S et al. The epidemiology of *Plasmodium falciparum* and *Plasmodium vivax* in East Sepik Province, Papua New Guinea, pre- and post-implementation of national malaria control efforts. Malar J. 2020; 19(1):198. https://doi.org/10.1186/s12936-020-03265-x PMID: 32503607
- Garfield RM, Vermund SH. Changes in malaria incidence after mass drug administration in Nicaragua. Lancet. 1983; 2(8348):500–3. https://doi.org/10.1016/s0140-6736(83)90523-8 PMID: 6136655
- McGreevy PB, Dietze R, Prata A, Hembree SC. Effects of immigration on the prevalence of malaria in rural areas of the Amazon basin of Brazil. Memorias do Instituto Oswaldo Cruz. 1989; 84(4):485–91. https://doi.org/10.1590/s0074-02761989000400005 PMID: 2487445
- Kondrashin AV, Sanyal MC. Mass drug administration in Andhra Pradesh in areas under *Plasmodium falciparum* containment programme. J Communic Dis. 1985; 17(4):293–9.
- 70. Landier J, Parker DM, Thu AM, Lwin KM, Delmas G, Nosten FH. Effect of generalised access to early diagnosis and treatment and targeted mass drug administration on *Plasmodium falciparum* malaria in Eastern Myanmar: an observational study of a regional elimination programme. *Lancet*. 2018; 391 (10133):1916–26. https://doi.org/10.1016/S0140-6736(18)30792-X PMID: 29703425
- 71. World Health Organization. Guidelines for the treatment of malaria. World Health Organization; 2015. Report No.: 978 92 4 154912 7.

- 72. Hofmann NE, Karl S, Wampfler R, Kiniboro B, Teliki A, Iga J et al. The complex relationship of exposure to new Plasmodium infections and incidence of clinical malaria in Papua New Guinea. *eLife* 2017; 6: e23708 https://doi.org/10.7554/eLife.23708 PMID: 28862132
- 73. Robinson LJ, Wampfler R, Betuela I, Karl S, White MT, Li Wai Suen CS et al. Strategies for understanding and reducing the *Plasmodium vivax* and *Plasmodium ovale* hypnozoite reservoir in Papua New Guinean children: a randomised placebo-controlled trial and mathematical model. *PLOS Med* 2015; 12 (10):e1001891. https://doi.org/10.1371/journal.pmed.1001891 PMID: 26505753
- 74. Kligler I, Mer G Periodic intermittent treatment with chinoplasmine as a measure of malaria control in a hyperendemic area. *Revista di Malariologia* 1931; 10(4).
- 75. van DW. Mass treatment of malaria with chloroquine. Results of a trial in Inanwatan. *Trop Geogr Med* 1961; 13:351–6. PMID: <u>13886375</u>
- 76. Lwin KM, Imwong M, Suangkanarat P, Jeeyapant A, Vihokhern B, Wongsaen K et al. Elimination of *Plasmodium falciparum* in an area of multi-drug resistance. *Malar J* 2015; 14:319. <u>https://doi.org/10.1186/s12936-015-0838-5 PMID: 26275909</u>
- 77. Baird JK. 8-aminoquinoline therapy for latent malaria. *Clin Microbiol Rev* 2019; 32(4). https://doi.org/10. 1128/CMR.00011-19 PMID: 31366609
- Gabaldon A, Guerrero L. An attempt to eradicate malaria by the weekly administration of pyrimethamine in areas of out-of-doors transmission in Venezuela. *Am J Trop Med Hyg* 1959; 8(4):433–9.
- 79. Singh N, Shukla MM, Mishra AK, Singh MP, Paliwal JC, Dash AP. Malaria control using indoor residual spraying and larvivorous fish: a case study in Betul, central India. *Trop MedInt Health* 2006; 11 (10):1512–20.
- Seyoum D, Kifle YG, Rondeau V, Yewhalaw D, Duchateau L, Rosas-Aguirre A et al Identification of different malaria patterns due to *Plasmodium falciparum* and *Plasmodium vivax* in Ethiopian children: a prospective cohort study. *Malar J* 2016; 15:208. <u>https://doi.org/10.1186/s12936-016-1253-2</u> PMID: 27075667
- Doke PP, Sathe RS, Chouhan SP, Bhosale AS. Impact of single round of indoor residual spray with lambda-cyhalotrin 10% WP on *Plasmodium falciparum* infection in Akola district, Maharashtra State. J Communic Dis 2000; 32(3):190–200.
- Charlwood JD, Alecrim WD, Fe N, Mangabeira J, Martins VJ. A field trial with lambda-cyhalothrin (ICON) for the intradomiciliary control of malaria transmitted by Anopheles darlingi root in Rondonia, Brazil. Acta Trop 1995; 60(1):3–13. https://doi.org/10.1016/0001-706x(95)00092-s PMID: 8546036
- 83. Mishra AK, Bharti PK, Kareemi TI, Chand SK, Tidgam AS, Sharma RK et al. Field evaluation of zero vector durable lining to assess its efficacy against malaria vectors and malaria transmission in tribal areas of the Balaghat district of central India. *Trans Roy Soc Trop Med Hyg* 2019; 113(10):623–31. https://doi.org/10.1093/trstmh/trz046 PMID: 31251386
- Rowland M, Hewitt S, Durrani N, Bano N, Wirtz R. Transmission and control of vivax malaria in Afghan refugee settlements in Pakistan. *Trans Roy SocTrop Med Hyg* 1997; 91(3):252–5. https://doi.org/10. 1016/s0035-9203(97)90065-1 PMID: 9231188
- Rowland M, Mahmood P, Iqbal J, Carneiro I, Chavasse D. Indoor residual spraying with alphacypermethrin controls malaria in Pakistan: a community-randomized trial. *Trop Med Int Health*. 2000; 5(7):472– 81. https://doi.org/10.1046/j.1365-3156.2000.00581.x PMID: 10964269
- Metselaar D. Seven years' malaria research and residual house spraying in Netherlands New Guinea. Am J Trop Med Hyg 1961; 10:327–34. https://doi.org/10.4269/ajtmh.1961.10.327 PMID: 13769879
- 87. Rowland M, Hewitt S, Durrani N. Prevalence of malaria in Afghan refugee villages in Pakistan sprayed with lambdacyhalothrin or malathion. *Trans Roy Soc Trop Med Hyg* 1994; 88(4):378–9.
- Sharma SN, Shukla RP, Raghavendra K, Subbarao SK. Impact of DDT spraying on malaria transmission in Bareilly District, Uttar Pradesh, India. J Vector Borne Dis. 2005; 42(2):54–60. PMID: 16161701
- Auburn S, Cheng Q, Marfurt J, Price RN. The changing epidemiology of *Plasmodium vivax*: Insights from conventional and novel surveillance tools. *PLOS Med* 2021; 18(4):e1003560. https://doi.org/10. 1371/journal.pmed.1003560 PMID: 33891580
- World Health Organization. Malaria 2020 [updated 14.1.2020. Available from: https://www.who.int/ news-room/fact-sheets/detail/malaria.
- Ashton RA, Prosnitz D, Andrada A, Herrera S, Yé Y. Evaluating malaria programmes in moderate- and low-transmission settings: practical ways to generate robust evidence. *Malar J* 2020; 19(1):75. <a href="https://doi.org/10.1186/s12936-020-03158-z">https://doi.org/10.1186/s12936-020-03158-z</a> PMID: 32070357